Author's response to reviews

Title: HER2 overexpression and amplification is present in a subset of ovarian mucinous carcinomas and can be targeted with trastuzumab therapy

Authors:

Jessica N McAlpine (mcalpine_j@yahoo.com)
Kimberly C Wiegand (KWiegand@cw.bc.ca)
Russell Vang (rvang@jhmi.edu)
Brigitte M Ronnett (bronnett@jhmi.edu)
Anna Adamiak (aadamiak2005@meds.uwo)
Martin Koebel (martin.koebel@vch.ca)
S E Kalloger (skalloger@mac.com)
Kenneth D Swenerton (KSwener@bccancer.bc.ca)
David G Huntsman (dhuntsma@bccancer.bc.ca)
C Blake Gilks (blake.gilks@vch.ca)
Dianne M Miller (dianne.miller@vch.ca)

Version: 5 Date: 4 November 2009

Author's response to reviews:

I will update the manuscript with a section declaring COI to include the following statement:

Martin Kobel was supported through a non directed educational grant from Eli Lilly Canada.